Author's response to reviews

Title: Patterns of comorbidity in community-dwelling older people hospitalised for fall-related injury: A cluster analysis

Authors:

Trang Vu (trang.vu@monash.edu)
Caroline F Finch (caroline.finch@monash.edu)
Lesley Day (lesley.day@monash.edu)

Version: 6 Date: 18 August 2011

Author's response to reviews:

Figure 2 and Figure 3 have been revised to include the title of the Y axis which was cropped in the previous version.